Multitask GANs for Oil Spill Classification and Semantic Segmentation Based on SAR Images

Jianchao Fan[®], *Member*, *IEEE*, and Chuan Liu

Abstract—The increasingly frequent marine oil spill disasters have great harm to the marine ecosystem. As an essential means of remote sensing monitoring, synthetic aperture radar (SAR) images can detect oil spills in time and reduce marine pollution. Many look-alike oil spill regions are difficult to distinguish in SAR images, and the scarcity of real oil spill data makes it difficult for deep learning networks to train effectively. In order to solve the abovementioned problems, this article designs a multitask generative adversarial networks (MTGANs) oil spill detection model to distinguish oil spills and look-alike oil spills and segment oil spill areas in one framework. The discriminator of the first generative adversarial network (GAN) is transformed into a classifier, which can effectively distinguish between real and look-alike oil spills. The generator of the second GAN model integrates a fully convolutional symmetric structure and multiple convolution blocks. Multiple convolution blocks can extract the shallow oil spill information, and the fully convolutional symmetric structure can extract the deeper features of the oil spill information. The algorithm only needs to use a small number of oil spill images as the training set to train the network, and the limitation of the oil spill dataset can be solved. Validation evaluations are conducted on three datasets of Sentinel-1, ERS-1/2, and GF-3 satellites, and the experimental results demonstrate that the proposed MTGANs oil spill detection framework outperforms other models in oil spill classification and semantic segmentation. Among them, the classification accuracy of the oil spill and look-alikes can reach 97.22%. The average OA for semantic segmentation of the oil spill area can be 97.47% and the average precision can reach 86.69%.

Index Terms—Generative adversarial networks (GANs), multitask, oil spill image classification, semantic segmentation, synthetic aperture radar (SAR).

I. INTRODUCTION

T HE reasons for the occurrence of marine oil spills are caused mainly ship accidents, marine transportation, and

Manuscript received 7 November 2022; revised 26 December 2022 and 28 January 2023; accepted 13 February 2023. Date of publication 27 February 2023; date of current version 16 March 2023. This work was supported in part by the National Natural Science Foundation of China under Grant 42076184, Grant 41876109, and Grant 41706195, in part by the National Key Research and Development Program of China under Grant 2021YFC2801000, in part by the National High Resolution Special Research under Grant 41-Y30F07-9001-20/22, and in part by the Dalian High Level Talent Innovation Support Plan under Grant 2021RD04. (*Corresponding author: Jianchao Fan.*)

Jianchao Fan is with the School of Control Science and Engineering, Dalian University of Technology, Dalian 116023, China, and also with the Department of Ocean Remote Sensing, National Marine Environmental Monitoring Center, Dalian 116023, China (e-mail: fjchaonmemc@163.com).

Chuan Liu is with the School of Information, Shanghai Ocean University, Shanghai 201306, China (e-mail: m210911486@st.shou.edu.cn).

The code of this work will be available at https://github.com/fjc1575/Marine-Oil-Spill/tree/main/MTGANs for the sake of reproducibility.

Digital Object Identifier 10.1109/JSTARS.2023.3249680



Fig. 1. Illustration of oil spill detection process based on SAR images.

the leakage of offshore oil drilling platforms. The threat of marine oil spills to the marine water ecosystem is more severe than other accidents [1]. Once an oil spill occurs in the ocean, it will not only damage the marine environment but also harm marine life. The use of satellite remote sensing technology to monitor marine oil spills can detect marine oil spills in time, deal with them quickly after disasters occur, and better reduce marine pollution. Among many remote sensing technologies, synthetic aperture radar (SAR) can observe the ground on all-day and has the sensitivity of oil film pressure wave effect, which has become an essential means of oil spill monitoring [2], [3]. The polarization mode of SAR image will affect oil spill detection. The VV polarization from the sea surface exhibits stronger backscattering than the HH polarization, and VV polarized images are much brighter than VH images due to their strong scattering. Thus, oil spills can be better detected using VV polarized SAR images [4]. The process of detecting oil spills based on SAR images is shown in Fig. 1. Marine oil spills appear in dark areas of SAR images due to the specular backscattering [5]. However, there will be a lot of look-alikes areas, such as upwelling, low wind speed areas, leeward areas, and biological slicks, also appearing as dark regions in SAR images [6]. Therefore, a crucial task is to correctly classify these dark areas into real or look-alike oil spills in the beginning.

Due to the development, the machine learning has been used by more and more researchers to solve many complex problems. Li et al. [7] used the support vector machine (SVM) model to detect oil slicks and similar objects with high accuracy. However, when using SVM for oil spill classification, it is necessary to

This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/

consider selecting an appropriate kernel function and its parameter configuration. Tong et al. [8] utilized the random forest to detect marine oil spills on polarimetric SAR data. Bandiera et al. [9] proposed a Bayesian edge detector that can be used to detect black spots on the ocean surface and, therefore, can be used as a first stage in identifying and monitoring oil spills. Basic image processing techniques, such as K-means clustering [10] to detect oil spills. Frate et al. [11] used neural networks for oil spill detection on ERS-SAR data. Li et al. [12] applied four types of machine learning models to study oil spill identification or classification and described the reflectance characteristics. The results obtained were consistent with those of previous studies. For the classical machine learning methods, feature extraction and selection of oil spills are carried out manually. These depend on a lot of expert experience and take a long time to process.

In recent years, deep learning neural networks have been applied to oil spill detection due to the advantages of automatical feature extraction and high detection precision [13], [14], [15]. Chen et al. [16] used two deep learning algorithms to identify oil spills and classify them as mineral, emulsion, and natural oil slicks. Yekeen et al. [17] developed a new deep learning oil spill detection model based on the mask region-based convolutional neural network model, in which, even in the case of overlapping, the target could be segmented with high accuracy. Zhang et al. [18] proposed a supervised detection network for petroleum context and boundary (CBD-Net) that fused multiscale features to extract oil spill areas. The performance of the CBD-Net network had the great improvement compared with the U-Net [19] model. Zhou et al. [20] proposed U-Net++ based on U-Net. Ma et al. [21] proposed an intelligent oil spill detection architecture based on a deep convolutional neural network, which performed better than traditional methods. Laurentiis et al. [22] studied oil slick classification under the framework of deep learning, and this classifier could separate mineral oil film from biological oil slick and clean ocean. Nieto-hidalgo et al. [23] proposed a ship and oil spill detection system based on airborne side-looking radar images. This method was composed of three pairs of convolutional neural networks, and the results had shown that the method was effective. Abdul et al. [24] studied the application of deep learning in oil spill detection and classification and modified U-Net for oil spill and similarity recognition. Xia et al. [25] proposed an SAR image segmentation method based on the principle of nonlocal processing with a multiscale active contour model, and the results displayed the effectiveness and feasibility of the method. Raeisi et al. [26] used an efficient cuckoo search algorithm and nonnegative matrix factorization of different Zernike moment features for discrimination between oil spills and look-alikes in SAR images. Aghaei et al. [27], [28] developed GreyWolfLSM and OSDES-Net methods for oil spill detection. Wang et al. [29] used the convolutional neural network AlexNet model to perform local connection, weight sharing, and learning representation of oil spill images to extract oil spill information from SAR images. Wang et al. [30] presented the MDOAU-Net model for SAR image segmentation in aquaculture raft monitoring. Two novel fusion networks were proposed for hyperspectral and SAR image classification [31], [32]. Deep learning methods [33], [34] achieved good performance in image segmentation, but there are still limitations. Most deep learning relies on a large amount of training data to ensure the accuracy of detection results, while in reality, there is a relative lack of a large amount of actual oil spill data.

To sum up, achieving accurate identification and detection is a problem in the case of limited samples. Zhang et al. [35] offered a graph information aggregation cross-domain few-shot learning framework for image classification. So far, some methods have been developed, such as metalearning [36], [37], metric learning [38], [39], transfer learning [40], [41], these methods mainly make full use of existing knowledge and experience to guide the learning progress of new tasks, but these methods rely on the feature representation ability of the model, which is complex and unintuitive. A straightforward way to solve the limited-sample learning problem is to increase the number of training samples through data augmentation. However, manual design is required in all of these methods [42]. As another data augmentation method, sample generation can solve the abovementioned problems. Sample generation cannot only expand the number of samples but also enrich the content of samples. These generated samples can improve the model is generalization ability and suppress the risk of overfitting. Generative adversarial networks (GANs) are an emerging type of generative model. GANs [43] were first proposed in 2014. Because the original GAN training is not constrained, Mirza et al. [44] improved the network and proposed conditionalGANs to guide the generator to synthesize the fake samples. Radford et al. [45] designed deep convolution generative adversarial networks, combining convolutional neural networks and GANs, effectively achieving high-quality image generation. Augustus et al. [46] offered semisupervised learning with generative adversarial networks (SGAN), an efficient classifier created. Gulrajani et al. [47] used Wasserstein GAN (WGAN) to solve the training instability problem. Later, an improved Wasserstein GAN-gradient penalty (WGAN-GP) [48] was presented based on the WGAN scheme. With the gradual development of GANs, the application fields have become more extensive, not only in data generation but also in image segmentation [49], [50], image classification [51], [52]. Other fields also have very amazing performance. GANs can expand the number of samples and increase adversarial samples to improve the ability of discriminative generalization. Only a small number of training samples are required during training. Therefore, GANs can also be used as a method for oil spill detection, Yu et al. [53] used a detection model with adversarial f-divergence learning for automatic oil spill identification. Li et al. [54] developed a multiscale conditional adversarial network (MCAN) using the GAN network. Only four oil spill images are needed in the network training, and the limitation that a large number of oil spill data are difficult to obtain can be solved. But it does not deal well with a variety of SAR images with small oil leak proportions.

Note that these GAN models do not consider how to distinguish real oil spills from look-alike oil spills, which is the most important part of oil spill monitoring. Moreover, it is hard for GANs to converge effectively because it is difficult to generate good adversarial samples. So, we make full use of these look-alike oil spills as effective adversarial samples to improve the classification and segmentation accuracy in the adversarial learning process, which can effectively combine the advantages of the GAN model with the special application of oil spills. In addition, the previous oil spill detection was a single task. It is not yet possible to simultaneously achieve the classification and segmentation of oil spills in one framework. How to design an end-to-end structure is more in line with the actual daily monitoring of oil spill satellite remote sensing.

Therefore, this article proposes a multitask generative adversarial networks (MTGANs) oil spill detection model to distinguish real and look-alike oil spills and segment real oil spill areas in one framework. The main contributions of this study are as follows.

- Design an MTGANs oil spill detection framework. The two GANs network models can accurately distinguish between real and look-alike oil spills, and the oil spill area can be segmented simultaneously for the first time.
- 2) Modify the discriminator of GANs. The discriminator is trained as an oil spill multiclassifier to discriminate real oil spill images, look-alike oil spill images, and pseudosamples accurately, while the generator is used to enrich similar samples to improve the discriminative ability of the model.
- 3) Integrate a fully convolutional symmetric structure and multiple convolution blocks in the oil spill semantic segmentation task. Multiscale oil spill features are extracted, respectively, to improve semantic segmentation accuracy with very little training data.

The rest of this article is organized as follows. Section II introduces the related work related to the research direction of this article and provides prior knowledge. Section III introduces the proposed MTGANs model in detail for SAR image oil spill. Section IV shows the experimental results of the work. Finally, Section V summarizes the highlights of the article and future research.

II. RELATED WORKS

A. Generative Adversarial Network

In 2014, Goodfellow et al. [43] first proposed GAN. It marks the official birth of GAN. In the original network, the GAN is a network composed of two models, one is the generator, and the other is the discriminator. The purpose of the generator is to learn the data distribution of real images and to try to generate real images to deceive the discriminator. The discriminator's goal is to distinguish the images generated by the generator from the real images. The generator and the discriminator constitute a dynamic game process. Eventually, in the ideal state, a Nash equilibrium state is reached. The generator generates enough fake images to fool the discriminator. For the discriminator, it is difficult to determine whether it is from the real data or the data generated by the generator. The GAN model framework is shown in Fig. 2. The generator inputs random noise z and generates pseudosamples G(z), then G(z) are fed into the discriminator together with the real data, and true or false is output by the discriminator. The output of the discriminator will guide the



Fig. 2. GAN framework.



Fig. 3. SGAN framework.

generator in reverse so that the generator can generate more realistic images.

The traditional GAN network is a simple two-layer feedforward neural network, which is not combined with the convolutional neural network and cannot effectively achieve highquality image generation. The generated image samples are also unconstrained, making predicting the results difficult. Therefore, the oil spill segmentation task cannot be performed directly.

B. Semisupervised Generative Adversarial Network

So far, the goal of most GAN variants is to generate realistic data samples. Therefore, generators are always a concern. The main purpose of the discriminator is to help the generator to improve the quality of the generated images. At the end of the training, discriminators are usually ignored and only trained generators are used to generate real data.

The main concern in SGAN [46] is the discriminator, and the SGAN combines the discriminator and the multiclassifier into one. The discriminator here does not just need to distinguish between true and false but learns to distinguish n + 1 classes, where n is the number of classes in the training dataset, and 1 adds a class to the pseudosamples generated by the generator, the original binary classifier is transformed into a multiclassifier. The goal of the discriminator during the training process is to distinguish between classes. The purpose of the generator is to improve its classification accuracy by providing additional information to help the discriminator learn relevant patterns in the data. At the end of the training, the generator will be discarded, and the trained discriminator will be used as the classifier. The model framework of SGAN is shown in Fig. 3. The generator inputs random noise z and outputs the generated pseudosamples G(z). The input of the discriminator consists of three parts, the data with class labels, the data without class



Fig. 4. MTGANs framework. In the design of the MTGANs oil spill detection framework, the input of the whole framework is the images of oil spills and look-alikes, and the final output is the generated oil spill segmentation map. The framework is divided into three parts in total. (I) Classification and identification of oil spill images. (II) Oil spill image judgment. (III) Semantic segmentation of oil spill images.

labels, and the pseudosamples generated by the generator. After the discriminator is discriminated, the output is the classification result.

Therefore, SGAN can be used in the research field of oil spill classification and recognition. A discriminator is trained to achieve the classification of oil spills and look-alike spills.

III. PROPOSED METHOD

A. Overview

The model consists of three parts, as shown in Fig. 4. The first part has a generator and a discriminator. The generator's input is random noise, and the generator outputs pseudosamples. Thermal noise removal, coherent speckle filtering, and terrain correction are preprocessed for the oil spill image and look-alike oil spill image. The preprocessed image is fed to the discriminator along with the pseudosamples generated by the generator. After input discriminator, the discriminator realizes the oil spill classification, look-alike, and fake sample classification. The role of the second part is the connection between the first and third parts. It is screened by the results of the classification in the first part, and the first part is classified as an image of the oil spill, and its label value is compared. The groundtruth label is the image of the oil spill, and the filtered result is used as the input for the next part. The third part is composed of a generator and a discriminator. The generator is input results from the screening in the second part, and the output is the generated oil spill segmentation map. The real oil spill image, the generated segmentation map, and the segmentation label map

are fed into the discriminator. The discriminator distinguishes the generated segmentation map from the real segmentation map. Reverse optimization of the generator according to the discriminant results so that the generator can generate a more realistic oil spill segmentation map and realize the semantic segmentation of oil spill images.

B. Oil Spill Classification Model

The oil spill classification model is built according to the idea of an SGAN. The classification model is shown in the first stage of Fig. 4. Use to classify oil spills and the like. The classification model consists of a generator G_c and a discriminator D_c , the discriminator is improved into a classifier for the oil spill classification task. The construction of the generator starts with the BatchNorm layer. The BatchNorm layer can solve the problem that the data distribution changes during the training process so as to prevent the gradient from disappearing or exploding and speeding up the training speed. Then there are upsampling layers, convolutional layers, BatchNorm layers, and LeakyReLU activation function. The function of the upsampling layer is to increase the length and width of the image and then input the convolution layer. To prevent vanishing gradients and overfitting during feature extraction, a normalization layer is added after the convolution layer. The LeakReLU activation function is added after the BatchNorm layer, which can make the network converge quickly. Then it also goes through the upsampling layer, the convolution layer, the BatchNorm layer, and a BatchNorm layer, and the LeakyReLU

Algorithm 1: Classification model.
Input: x : real and look-alike oil spill images
y : classification label
z : random noise
CE : cross entropy loss
BCE : binary cross entropy loss
Output: Classification result
for epochs do
$x' = G_c(z, y)$
Loss $D_c = \operatorname{CE}(D_c(x', y) + D_c(x, y)) +$
$BCE(D_c(x') + D_c(x))$
Update D_c
Loss $G_c = CE(D_c(x', y)) + BCE(D_c(x'))$
Update G_c
end for

activation function in sequence. Finally, the convolutional layer and the Tanh activation function. The discriminator consists of four convolutional blocks, each of which has a convolutional layer, a BatchNorm layer, a LeakyReLU activation function connected in turn. The output layer has a sigmoid activation function and a softmax activation function. The sigmoid function is used to distinguish between true and false, after the output of the sigmoid function, there will be two cases representing the true or false samples. The softmax function is used for multiclass classification. After the softmax function output, the discrimination between look-alike oil spill images, oil spill images, and fake samples can be achieved.

The generator and discriminator of the oil spill classification model need to input oil spill category information corresponding to the training data during the training process. Therefore, the oil spill classification model is a supervised model. The input to the generator is random noise z, and the output of the generator is the generated pseudosample. The discriminator inputs the pseudosample generated by the generator, the oil spills, and look-alike oil spills marked by the real category. The crossentropy (CE) loss function and binary cross-entropy (BCE) loss function are used to calculate the loss of discriminant results in classification models. CE is used after the softmax activation function and defined as

$$CE = -\sum_{i=1}^{n} y_i \log(f(x_i)) \tag{1}$$

where, $f(x_i)$ is the predicted value of the output of the sample x after passing through the network. y represents the true category information corresponding to the corresponding sample x. The CE function was used to calculate the loss of multiple classification results of the discriminator, which improved the accuracy of the oil spill, look-alike spill, and pseudosample classification of the model. BCE loss function is defined as

$$BCE = -b\log \hat{b} - (1-b)\log(1-\hat{b})$$
(2)

where, b is the true value of sample x, which is true or false. b is the value predicted by the network. BCE is used to distinguish between real samples and fake samples, and this loss function

guides the generator to generate images closer to the real sample. So, the discriminator can distinguish between real and fake, and classify real and look-alike oil spills. The objective function of the oil spill classification model is defined as

$$\min_{G_c} \max_{D_c} V(D_c, G_c) = E_{x \sim p_{data}(x)} \sum_{i=1}^n [y_i \log(D_c(x_i)) + y_i \log(D_c(x_i)) + \log D_c(x)] + E_{z \sim p_z(z)} [\log(1 - D_c(G_c(z)))]$$
(3)

where, z represents random noises, x represents real and lookalike oil spill images, y is oil spill category information, and $G_c(z)$ is the generated pseudosample. The classification model training strategy is formally described in Algorithm 1. The discriminator is trained to maximize the discrimination ability. With the improvement of the discriminator, the true and false samples are distinguished more correctly. The generator is trained to the minimization to generate images as close as possible to the real oil spill and look-alike oil spill, and finally, successfully fool the discriminator.

After the final training, the generator is discarded, and the discriminator is retained, through which the purpose of classifying the oil spill image and the imitation oil spill image can be achieved. On the other hand, the convolutional parts of (the classification part) and (the semantic segmentation part) have the same structure. The trained D_c weights are set to the D_s initial weights. The effect of similar adversarial images is transferred to the semantic segmentation part to improve the convergence speed and segmentation accuracy of the algorithm. Therefore, the same structural parts of the two GANs models share weight information in the multitask framework.

C. Oil Spill Image Judgment

The judgment model is shown in the second stage of Fig. 4. Because the oil spill images identified in the previous step may have misclassifications of look-alikes oil spill images, to ensure that they are all oil spill images during semantic segmentation, a screening operation should be performed to eliminate the oil spill images misclassified as oil spill images. Here, the label value is used for the judgment operation. The label values are the oil spill and look-alike oil spill category labels in the first stage. If the true label value is also an oil spill image according to the discrimination result, this image will be retained. Otherwise, the image is rejected. After filtering, the data left are all oil spill images, which will be used as input for the next step.

D. Semantic Segmentation Model

The segmentation model is shown in the third stage of Fig. 4, and consists of the generator G_s and the discriminator D_s . The generator network first consists of four downsampled and four up-sampled fully convolutional symmetric structure networks, forming a U-shaped structure. The pooling layer and the double-layer convolutional layer are the downsampling components, and two convolutional layers and a ReLU activation function

Algorithm 2: Segmentation model.
Input: <i>I</i> : oil spill images
S : segmentation label
Output Generated oil spill segmentation map \widehat{S}
for epochs do
$L_{D_s} =$
$D_s(I,\widehat{S}) - D_s(I,S) + \lambda_2 (\nabla_{\widehat{S}} D_s \ S - \widehat{S}\ _2 - 1)^2$
Update D_s
$L_{G_s} = \lambda_1 \ S - \widehat{S}\ _1 - D_s(I, \widehat{S})$
Update G_s
end for

form the double-layer convolutional layer. Upsampling consists of deconvolutional layers and double-layered convolutional layers. Two convolutional layers and a ReLU activation function are combined into two convolutional layers. For the four times of fully convolutional symmetric structure sampling, the image is scaled to a total of 16 times. Symmetrically, the upsampling is performed four times, and the obtained high-level semantic feature map is restored to the resolution of the original image. After four times of upsampling, the information, such as the edge of the oil spill segmentation map is also recovered more finely. The fully convolutional symmetric structure is added to the generator, which can extract the feature information of the oil spill from global to local and generate a more accurate oil spill segmentation map with the help of the discriminator. The fully convolutional symmetric structure network is followed by a convolutional layer and three convolutional blocks, each having a convolutional layer, a BatchNorm layer, and a LeakyReLU activation function. Finally, there is a convolutional layer and a Tanh activation function. The construction of the discriminator consists of four convolutional blocks, each containing a convolutional layer, a BatchNorm layer, and a LeakyReLU activation function. The last layer is a convolutional layer.

The segmentation model is trained using WGAN-GP loss [48] to stabilize the training. The objective function of the generator is shown in the following:

$$L_{G_s} = \lambda_1 \| S - \hat{S} \|_1 - D_s(I, \hat{S})$$
(4)

where, λ_1 is the G_s weight balance parameter, S is the input real segmentation label map, \hat{S} is the segmentation map generated by the generator, and I is the real oil spill image. $||S - \hat{S}||_1$ is the l_1 norm, which penalizes the pixelwise distance between the segmentation label map and the generated segmentation map. Minimizing the training of the generator to generate a more realistic segmentation map of the oil spill can finally deceive the discriminator successfully.

The objective function of the discriminator is shown in the following:

$$L_{D_s} = D_s(I, \widehat{S}) - D_s(I, S) + \lambda_2 \left(\nabla_{\widehat{S}} D_s \| S - \widehat{S} \|_2 - 1 \right)^2$$
(5)

where, λ_2 is D_s the weight balance parameter, this item $D_s(I, \widehat{S}) - D_s(I, S)$ represents the adversarial loss, which increases the discriminant's discriminative ability. The term

 $\lambda_2 (\nabla_{\widehat{S}} D_s || S - \widehat{S} ||_2 - 1)^2$ represents penalized gradient loss, which produces stable gradients that neither vanish nor explode. \widehat{S} represents a random variable sampled uniformly between Sand \widehat{S} . Minimize the training of the discriminator to distinguish the generated oil spill segmentation map from the oil spill segmentation label map.

The generator input is the real oil spill image, and the output is the semantic segmentation map of the oil spill image. The input to the discriminator is the real oil spill image, the semantic segmentation map generated by the generator, and the oil spill image segmentation label map. After being discriminated by the discriminator, the discriminant score is output. The discriminator then reverse optimizes the generator to guide the generator to generate a more accurate oil spill segmentation map. In the initial training of D_s , the D_c weights trained in the classification model are loaded, and the effect of similar adversarial images is transferred to the semantic segmentation part to improve the convergence speed and segmentation accuracy of the algorithm. At the end of the final training, the discriminator is discarded, and the generator is retained. The generator can generate an oil spill segmentation map of oil spill images. The purpose of semantic segmentation of the oil spill image is achieved. The segmentation model training strategy is formally described in Algorithm 2, the overall objective function of the segmentation model is shown in the following:

$$\min_{G_s} \max_{D_s} L(D_s, G_s) = D_s(I, S) - D_s(I, S) + \lambda_1 \|S - \widehat{S}\|_1 - \lambda_2 \left(\nabla_{\widehat{S}} D_s \|S - \widehat{S}\|_2 - 1 \right)^2.$$
(6)

After training, the segmentation model discards the discriminator and keeps the generator. The generator is used to generate the oil spill segmentation map to realize the segmentation task of the oil spill area. Finally, the discriminator of the classification model and the generator in the segmentation model are integrated to form an MTGANs, which is used for oil spill classification and semantic segmentation to form a complete oil spill detection framework. The process of the overall framework is the sequential learning method. After the classification and judgement part is completed, begin the semantic segmentation process. So learning errors in the semantic segmentation part are not backward to the judgment and classification models. However, the classification and segmentation models share the weights of the discriminator, whose convolutional parts have the same structure. D_c weights after training are set as D_s initial weights. The effect of similar adversarial images is transferred to semantic segmentation, which improves the convergence speed and segmentation accuracy. Therefore, the overall framework is not entirely multistage independent learning.

IV. EXPERIMENTS

A. Dataset Descriptions and Preprocess

In order to prove the effectiveness of the proposed method, the Sentinel-1, GF-3, and ERS-1/2 satellite data are used to conduct comparative experiments.

- 1) Sentinel-1 data: Two satellites of the Sentinel-1 satellite carry SAR sensors, active microwave remote sensing satellites with a resolution of up to 5 m and a width of 400 km. Interferometric wide swath imaging mode is used, the polarization mode is VV polarization, and the angle of incidence is 20° -45°. In the experiment, the observed oil spill data are selected and cut after preprocessing. Cropped oil spill image size 256×256 pixels.
- 2) *GF-3 data:* The GF-3 satellite is China's multipolarization SAR satellite. It was launched in 2016, and the sensor was equipped with C-band. The width is 10–650 km, and the resolution can reach 1 m. The imaging mode is FSI, FSII, and SS. The angle of incidence of FSI and FSII is $19^{\circ}-50^{\circ}$, and the incidence angle of SS is $17^{\circ}-50^{\circ}$. The polarization mode is VV polarization. The oil spill data observed in the experiment are preprocessed and cropped, and the oil spill image size is 256×256 pixels.
- 3) *ERS-1/2 data:* ERS-1 and ERS-2 ESA were launched in 1991 and 1995, respectively, with sensors equipped with C-band. The polarization mode is VV polarization, and the angle of incidence is $23^{\circ}-60^{\circ}$, and the width is 80–100 km. The observed oil spill data are selected and cut after preprocessing. Cropped oil spill image size 256×256 pixels.

Preprocessing operations of thermal noise removal, radiometric calibration, coherent scatter filtering, and terrain correction are performed on the acquired SAR images. Thermal noise removal performing thermal noise removal can improve the thermal noise problem. Radiation calibration due to the penetrating nature of clouds, a radiation calibration operation needs to be done on SAR data. Radiation calibration can eliminate the sensor's error and determine the accurate radiation value at the entrance of the sensor. Coherent spot filtering is a common phenomenon in SAR images. There are various filters to remove coherent spots. Here, the refined Lee filter is used. This is the most commonly used coherent spot filter. The processing effect is excellent. The refined Lee filter has a filter window size setting set to 7×7 window size. The terrain correction ensures that the real geographic coordinates correspond to the remote sensing image, and the distance Doppler method is used to correct the terrain. The image is input into the model after the abovementioned processing.

In the experiment, the groundtruths of oil spill SAR images are collected through relevant news reports, literature publications, and our daily oil spill monitoring. In addition, look-alike areas can be identified by querying the synchronous marine environment, such as wind speed, sea surface temperature, and chlorophyll concentration [6]. The semantic segmentation groundtruths of oil spill SAR images are generated by LabelMe annotation tool.

B. Experimental Setup

All experiments are compiled under Windows 10 with python 3.8, pytorch 1.7.1, and cuda 10.1, run with GeForce RTX 2080 Ti GPU, and the input data size is 256×256 SAR images.

Classification model train generator G_c and discriminator D_c using Adam optimizer with $b_1 = 0.5$ and $b_2 = 0.999$, network learning rate is 0.0002. Classification model train 300 epochs. The classification model uses Sentinel-1 data with 120 images. The ratio of the training set and test set is set to 7:3.

Segmentation model train the generator G_s and discriminator D_s using the Adam optimizer with $\beta_1 = 0.5$ and $\beta_2 = 0.999$, the network has a learning rate of 0.0005 and a minibatch size of 1. The balance parameter of the ℓ_1 -norm constraint is $\lambda_1 = 10$, and the gradient penalty weight for the WGAN-GP loss is $\lambda_2 = 10$. The segmentation model is trained for 50 epochs. The training set and test set of the segmentation model include Sentinel-1 data, GF-3 data, ERS-1/2 data, of which Sentinel-1 data, ERS-1/2 data select four oil spill images as for training, 20 images are used for testing, four oil spill images are selected as training, and five images are used for testing GF-3 data.

Evaluate the performance of MTGANs on SAR images and separate the oil spill classification and segmentation models for their comparative experiments. The classification model is compared with the K-nearest neighbor algorithm (KNN) [55], SVM [7], random forest classifier (RFC) [8], decision tree classifier (DTC) [12], and the Bernoulli naive Bayes classifier (NBC) [9]. The segmentation model is compared with MCAN [54], SegNet [33], U-Net++[20], MODAU-net [30], DeepLabv3+[34], respectively. To ensure the reliability of the experiments, all comparative experiments use the same training set as training, and the same performance indicators are used for evaluation.

C. Evaluation Criteria

The accuracy rate is used to evaluate the performance of the classification model, which is the proportion of the total sample where oil spills and look-alikes oil spill images are correctly classified. The accuracy rate is shown in the following:

$$Accuracy = \frac{a}{m} \tag{7}$$

where, *a* represents the number of correctly classified samples, and *m* represents the total number of samples. Calculate the OA, Precision, F1-score, Kappa, and MIoU based on the confusion matrix to evaluate the performance of the oil spill segmentation model. OA represents the proportion of the total correctly predicted in the generated oil spill segmentation map and defined in the following:

$$OA = \frac{T_P + T_N}{T_P + T_N + F_P + F_N}.$$
 (8)

Recall represents what fraction of the oil in the spill area has been detected and defined in the following:

$$R = \frac{T_P}{T_P + F_N}.$$
(9)

Precision represents what proportion of the oil spill area in the generated oil spill segmentation map is a real oil spill defined in the following:

$$P = \frac{T_P}{T_P + F_P}.$$
(10)



Fig. 5. After the discriminator D_c discriminates the result graph, (a), (b), (c) are discriminate as oil spill images, (d), (e), (f) are discriminated as look-alike oil spill images.

F1-score considers both precision and recall and defined in the following:

$$F1 = \frac{2 \cdot P \cdot R}{P + R}.$$
 (11)

MIOU represents each class's average intersection ratio and is often used as an evaluation index for semantic segmentation and defined in the following:

$$M = \frac{1}{2} \left(\frac{T_P}{T_P + F_P + F_N} + \frac{T_N}{T_N + F_N + F_P} \right).$$
(12)

Kappa is used for image consistency check and defined in the following:

$$K = \frac{OA - P_e}{1 - P_e} \tag{13}$$

where

$$P_{e} = \frac{(T_{N} + F_{P}) \times (T_{N} + F_{N}) + (T_{P} + F_{P}) \times (T_{P} + F_{N})}{N \times N}.$$
(14)

D. Experiment 1: Sentinel-1 Dataset

1) Oil Spill Classification: Conduct experiments in the oil spill classification model. The classification results for the six images are presented in Fig. 5. Among them, Fig. 5(a), (b), and (c) is judged as oil spill images, and Fig. 5(d), (e), and (f) distinguished as look-alike oil spill images.

To ensure the reliability of the method for oil spill classification model comparison experiments. The data of Sentinel-1 is used, and the ratio of the oil spill and look-alike oil spill is set to 1:1 and 1:2, respectively, in training set to conduct the experiment, and compare KNN [55], DTC [12], SVM [7], RFC [8], and plain NBC [9] methods. The experimental results are shown in Table I. From the experimental results, in the case of the 1:2 ratio of the oil spill and look-alike oil spill data, the models did not show excellent performance due to the

TABLE I PERFORMANCE METRICS RESULTS FOR DIFFERENT RATIO OF THE SENTINEL-1 DATASET UNDER SIX CLASSIFICATION METHODS

Ratio	Methods	Accuracy (%)
	KNN [55]	86.11
	DTC [12]	80.55
1:2	SVM [7]	83.33
	NBC [9]	58.33
	RFC [8]	86.11
	MTGANs	94.44
	KNN [55]	88.89
	DTC [12]	86.11
1:1	SVM [7]	86.11
	NBC [9]	63.89
	RFC [8]	91.66
	MTGANs	97.22

The bold values indicate that the result is optimal compared to other proposed methods.

unbalanced ratio of the oil spill and look-alike oil spill data. After balancing the dataset and adjusting the ratio of the oil spill and look-alike oil spill data to 1:1, the model performance of all six methods improved. The NBC model assumes that the attributes should be independent of each other, and when the number of attributes is relatively large or the correlation between attributes is large, and the NBC classification effect is not good, and the correlation between the oil spill and suspected oil spill data are large, so there are a large number of images misclassified in the NBC model in oil spill classification recognition. All four models, KNN, DTC, RFC, and SVM, show good performance. Since RFC uses an integrated algorithm, its own accuracy is better than most individual algorithms, so it has high accuracy in oil spill classification recognition. The MTGANs proposed in this article consist of a generator and a discriminator in the oil spill classification model. The discriminator learns the feature information of oil spill and look-alike oil spill images and learns to distinguish between oil spill images and look-alike oil spill images. The pseudosamples generated by the generator enrich the number of samples and improve the discriminator's discrimination ability. Therefore, this article proposes that MTGANs have the highest accuracy in an oil spill and look-alike oil spill classification recognition.

2) Oil Spill Segmentation: Experiment with the oil spill segmentation model, both the training set and the test set are Sentinel-1 data, only four oil spill images are used for training, and 20 oil spill images are used for testing. The same training set and test set are used to conduct comparative experiments with MCAN [54], U-Net++[20], MODAU-Net [30], DeepLabv3+[34], and SegNet [33] models, respectively, and the results are evaluated using five performance metrics.

The oil spill segmentation results are compared and analyzed, and the segmentation results of three oil spill images under five models are shown in Fig. 6. Most of the oil spill images selected in this experiment are oil spill images with relatively large oil spill areas and regular shapes. In using a small amount of dataset to train the model, U-Net++ also maintains excellent performance in this experiment because U-Net++ supports a small amount of data to train the model. Although SegNet, DeepLabv3+, and MCAN also perform well in this experiment,



Fig. 6. Experimental results of three oil spill test images from Sentinel-1 data on five comparison methods. (a), (b), and (c) are three different test images.

the segmentation accuracy is not as excellent as U-Net++. The lack of dataset leads to the relatively blurred edge information of the oil spill region in the SegNet method segmentation map. In contrast, the U-Net++ and MCAN methods perform better in the edge region, but the U-Net++ and MCAN methods may misclassify the sea surface as the oil spill region. The MODAU-Net method has optimized edge information compared with the SegNet method, but there are a large number of misjudgments in the prediction graph. DeepLabv3+ uses an encoder and decoder. The encoder provides high-level feature semantic information, and the decoder replies to boundary information step by step, which improves the segmentation effect while focusing on the boundary information. It makes the DeepLabv3+ method superior to the SegNet method in handling boundary information, but the SegNet method severely misclassifies the sea surface as an oil spill area. The MTGANs method proposed in this article integrates a complete convolutional symmetric structure and multiple convolutional blocks in the segmentation model. The multiple convolutional blocks can extract shallow oil spill information, and the full convolutional symmetric structure can extract deep features of oil spill information. The generator can extract global to local oil spill information by refining the edge information of the oil spill segmentation map with four times downsampling and four times upsampling, and together with the discriminator, it can play a guiding role in generating the oil spill segmentation map for the generator. Therefore, the accuracy of MTGANs proposed in this article is better than other models for oil spill segmentation. The five performance metrics of the three oil spill test images are compared, and the metrics results are shown in Table II, and the performance metrics in the table are consistent with our visual results. Therefore, the MTGANs proposed in this article have the best oil spill image segmentation performance.

The average performance indexes of the five methods on 20 oil spill test images are evaluated. The average performance indexes are shown in Table III. It can be seen from the table that the MTGANs proposed in this article is also the highest in OA, Precision, F1-score, Kappa, and MIoU performance index

TABLE II PERFORMANCE INDICATORS OF SENTINEL-1 THREE TEST IMAGES UNDER FIVE SEGMENTATION METHODS

Samp.	Methods	OA(%)	P(%)	R(%)	F1(%)	K(%)	M(%)
	U-Net++ [20]	95.30	81.09	70.85	75.63	51.34	65.95
	MODAU-Net [30]	94.20	74.26	72.15	73.19	46.39	63.44
(a)	SegNet [33]	91.88	66.47	70.36	68.36	36.80	58.75
	DeepLabv3+ [34]	94.02	73.39	71.59	72.48	44.98	62.77
	MCAN [54]	95.69	73.29	76.62	74.92	49.87	65.41
	MTGANs	97.28	86.99	74.57	79.72	59.50	70.48
	U-Net++ [20]	96.74	78.29	91.05	84.17	68.44	75.27
	MODAU-Net [30]	92.86	67.86	81.24	73.95	48.68	63.75
(b)	SegNet [33]	95.26	72.92	85.80	78.84	57.92	69.08
	DeepLabv3+ [34]	94.76	71.04	86.57	78.04	52.89	66.46
	MCAN [54]	94.36	68.02	82.41	74.53	49.66	64.69
	MTGANs	98.17	83.77	91.29	87.37	74.75	79.50
	U-Net++ [20]	97.15	72.86	86.53	79.11	58.35	69.82
	MODAU-Net [30]	96.13	68.68	82.99	75.16	50.63	65.74
(c)	SegNet [33]	95.35	66.39	80.31	72.69	45.88	63.33
	DeepLabv3+ [34]	94.26	63.30	86.23	73.01	38.01	59.69
	MCAN [54]	96.34	66.11	81.39	72.96	46.36	63.85
	MTGANs	98.58	78.89	88.88	83.59	67.20	74.98

The bold values indicate that the result is optimal compared to other proposed methods.

TABLE III AVERAGE PERFORMANCE INDEX OF FIVE SEGMENTATION METHODS FOR SENTINEL-1 DATA

Methods	OA(%)	P(%)	R(%)	F1(%)	K(%)	M(%)
U-Net++ [20]	96.54	80.46	80.79	81.61	60.28	71.21
MODAU-Net [30]	95.20	77.35	79.49	78.41	60.36	71.62
SegNet [33]	93.40	73.42	77.26	75.30	51.10	66.02
DeepLabv3+ [34]	93.98	73.53	80.39	76.81	51.47	65.93
MCAN [54]	92.60	72.40	77.90	75.05	51.40	66.29
MTGANs	97.36	86.67	81.13	83.81	67.74	74.35

The bold values indicate that the result is optimal compared to other proposed methods.

average. The Kappa coefficient is often used to test the spatial consistency of image classification. The mean value of Kappa is 65.74%, when a Kappa value of 61%–80% is considered to be highly consistent, which is consistent with the results of the visual interpretation. Therefore, the MTGANs method proposed in this article can achieve high accuracy even with a small training set.



Fig. 7. Experimental results of three oil spill test images from ERS-1/2 data on five comparison methods. (a), (b), and (c) are three different test images.

TABLE IV Average Performance Metrics of Segmentation Results for 18 Oil Spill Images After Classification

Evaluation criteria(%)	MTGANs
OA	97.47
Р	86.69
R	80.13
F1	83.19
K	66.45
М	74.71

The MTGANs proposed in this article can be used for both oil spill classification and semantic segmentation of oil spill images. The overall experiments are performed on Sentinel-1 data. Forty images are selected as testing; there are 20 oil spill images and 20 oil spill similar images. The test image is input into the MTGANs framework for experiments. First, it passes through the discriminator D_c of the classification model and outputs the classification results. The test results have an accuracy of 95% for the discrimination of oil spills and look-alike oil spill images. Then, the discrimination results are screened and filtered. Eighteen images are identified as oil spills, and the real labels are also oil spills. The generator G_s of the 18 oil spill image segmentation model is used for semantic segmentation. Generate oil spill maps and calculate the average performance index of 18 oil spill images. The average performance indicators are shown in Table IV.

Some of the current oil spill detection methods can only perform oil spill classification, some can only perform oil spill segmentation, and few can achieve both oil spill classification and experimental oil spill segmentation. Therefore, a comparative test of oil spill classification and segmentation is carried out to evaluate the performance.

E. Experiment 2: ERS-1/2 Dataset

An oil spill segmentation experiment is conducted on ERS-1/2 satellite data. Four oil spill images are also selected as the training set and the rest as the test set, and the experimental results are shown in Fig. 7. In terms of visual effect, the segmentation effect of oil spills with thin strip shapes is not too accurate,

compared with the MTGANs method proposed in this article, the segmentation similarity is relatively high.

Three images are selected from the ERS-1/2 test data for analysis, namely, the long strip-shaped oil spill image, the small area oil spill image, and the complex and elongated shape oil spill image. The experimental results of this method are shown in Fig. 7. Since most of the selected oil spill images in this experiment are small and irregularly shaped oil spill areas, although U-Net++ supports a small amount of data to train the model, the oil spill images of small areas will lead to training difficulties. Therefore, the segmentation performance of U-Net++ is low in this experiment, and the same situation occurs in SegNet and DeepLabv3+. From the three result images, the U-Net++, SegNet, and DeepLabv3+ methods can only segment the general shape of the oil spill, and the segmented oil spill area is not obvious, as shown in Fig. 7(b). The MODAU-Net method has excellent segmentation performance on ERS-1/2 dataset, but it will miss the edge information of oil spills. For the small area oil spill image, the shape of the whole oil spill area cannot be completely segmented in the segmentation result. The segmentation results for complex oil spill images cannot completely segment the shape of the entire oil spill area, while the slender shape of the oil spill cannot be segmented at all. Compared with the other three methods, MCAN outperforms U-Net++, SegNet, and DeepLabv3+ in segmenting oil spill images of small regions because it uses a multiscale strategy with cascaded coarse-to-fine data streams to enhance the representation of the model. However, MCAN can have incorrectly segmented areas, incorrectly segmenting areas that are not oil spills as oil spill areas. The MTGANs method proposed in this article is integrated with a fully convolutional symmetric structure and multiple convolutional blocks due to the integration in the generator. It can capture oil spill information from shallow to deep layers. Therefore, MTGANs can achieve oil spill image segmentation of small areas and reduce the misclassification of oil spill areas. The experimental results show that the MTGANs proposed in this article outperform other models in segmenting oil spill images with small areas and irregular shapes. The performance indexes of the three oil spill images are shown in Table V. From the data

TABLE V PERFORMANCE INDICATORS OF ERS-1/2 THREE TEST IMAGES UNDER FIVE SEGMENTATION METHODS

Samp.	Methods	OA(%)	P(%)	R(%)	F1(%)	K(%)	M(%)
	U-Net++ [20]	94.81	80.63	48.22	60.35	22.21	54.01
	MODAU-Net [30]	95.89	91.49	73.41	81.46	62.54	71.56
(a)	SegNet [33]	95.31	87.12	63.84	73.69	39.62	60.71
	DeepLabv3+ [34]	93.15	90.16	71.76	79.92	54.64	67.67
	MCAN [54]	95.61	80.12	77.02	78.54	56.95	68.79
	MTGANs	96.79	92.15	78.33	82.90	64.90	72.13
	U-Net++ [20]	97.08	91.95	66.09	74.13	52.42	60.47
	MODAU-Net [30]	97.14	94.34	63.41	73.66	52.91	63.44
(b)	SegNet [33]	96.83	69.02	60.13	64.27	26.32	56.47
	DeepLabv3+ [34]	94.48	87.09	63.83	66.54	13.63	52.42
	MCAN [54]	97.14	73.50	65.42	69.23	37.20	60.51
	MTGANs	97.96	96.86	71.54	77.47	56.95	66.36
	U-Net++ [20]	93.18	90.47	64.98	72.86	51.34	64.94
	MODAU-Net [30]	93.10	86.87	67.59	76.03	52.09	66.01
(c)	SegNet [33]	92.35	85.35	57.27	68.55	22.98	53.30
	DeepLabv3+ [34]	92.64	90.16	59.08	70.65	25.59	54.27
	MCAN [54]	93.11	79.77	67.20	72.36	41.45	60.87
	MTGANs	94.01	91.88	70.98	76.14	52.36	67.58

The bold values indicate that the result is optimal compared to other proposed methods.

TABLE VI Average Performance Index of Five Segmentation Methods for ERS-1/2 Data

Methods	OA(%)	P(%)	R(%)	F1(%)	K(%)	M(%)
U-Net++ [20]	94.04	76.01	58.78	66.30	33.46	58.89
MODAU-Net [30]	94.09	76.81	62.89	69.16	38.88	61.40
SegNet [33]	92.81	70.58	56.98	63.06	24.36	55.44
DeepLabv3+ [34]	72.35	67.91	45.72	54.66	20.97	43.07
MCAN [54]	90.41	76.18	69.56	72.72	41.93	60.36
MTGANs	95.51	85.59	70.04	76.93	59.13	65.48

The bold values indicate that the result is optimal compared to other proposed methods.

in the table, the performance index values of Precision, F1-score, Kappa, and MIoU are also the highest for the MTGANs method proposed in this article. The performance metrics in the table are consistent with our visual results.

Combined with the average performance index, the average performance index of 20 oil spill images is shown in Table VI. It can be seen from the table that the F1-score, Kappa, and MIoU index values of U-Net++, SegNet, and DeepLabv3+ methods are generally low because of the small data training set. Four training sets cannot guarantee the accuracy of prediction segmentation, and there will be inappropriate predicted images in the test results of U-Net++ and DeepLabv3+ methods. The similarity between the segmentation result map and the segmentation label map is almost 0, which is why the performance indicators of U-Net++ and DeepLabv3+ methods are low. Compared with them, the SegNet method does not have a complete prediction error, but the segmentation effect is not excellent. Both the MCAN method and the MTGANs method proposed in this article can solve the limitation of the lack of data and achieve good segmentation results with a small number of training sets. However, the similarity of the segmentation maps generated by the MCAN method is not high, and there are nonoil spill regions segmented into oil spill regions. Therefore, the MTGANs method proposed in this article has a better segmentation effect on the oil spill area.

TABLE VII PERFORMANCE INDICATORS OF GF-3 THREE TEST IMAGES UNDER FIVE SEGMENTATION METHODS

Samp.	Methods	OA(%)	P(%)	R(%)	F1(%)	<i>K</i> (%)	M(%)
	U-Net++ [20]	96.79	77.64	93.19	84.71	69.58	75.93
	MODAU-Net [30]	96.87	78.98	82.39	80.65	61.31	71.35
(a)	SegNet [33]	96.13	74.60	66.59	70.37	39.54	61.11
	DeepLabv3+ [34]	97.03	83.23	72.82	77.68	53.99	67.68
	MCAN [54]	97.69	82.29	92.42	88.37	75.25	79.72
	MTGANs	98.88	90.40	95.28	92.78	85.39	87.10
	U-Net++ [20]	65.72	51.24	35.56	41.99	22.21	33.96
(b)	MODAU-Net [30]	83.81	52.54	48.44	50.41	28.16	44.38
	SegNet [33]	97.64	57.83	68.77	62.83	22.00	55.31
	DeepLabv3+ [34]	98.42	65.73	82.91	73.33	42.42	62.95
	MCAN [54]	88.80	53.41	77.56	66.35	31.49	47.85
	MTGANs	98.53	67.82	84.18	75.12	55.82	64.42
	U-Net++ [20]	94.37	67.66	82.11	74.19	49.18	64.13
	MODAU-Net [30]	94.21	68.33	81.70	74.42	49.34	64.54
(c)	SegNet [33]	95.52	66.87	64.10	65.46	30.72	57.62
	DeepLabv3+ [34]	96.82	77.19	79.97	78.56	57.03	69.13
	MCAN [54]	96.19	74.70	91.94	82.43	61.69	71.31
	MTGANs	97.41	80.51	91.08	85.47	69.88	76.30

The bold values indicate that the result is optimal compared to other proposed methods.

F. Experiment 3: GF-3 Dataset

Due to the lack of a GF-3 dataset, nine oil spill datasets are selected for the oil spill segmentation experiment. Among the nine oil spill datasets, four oil spill images are used as the training set, and five oil spill images are used as the test set.

The test results of the three GF-3 oil spill images are shown in Fig. 8. Due to the lack of GF-3 data, there will be images with bad oil spill imaging and more noise interference in the dataset, and the U-Net++ method still has excellent segmentation performance for large area oil spill images, as shown in Fig. 8(a), which is also consistent with the results of Experiment 1. The segmentation performance is low for the images with bad oil spill imaging and more noise interference, as in Fig. 8(b). The MODAU-Net method has better segmentation performance than U-Net++ for complex oil spill images, but the segmentation consistency is low, and the oil spill cannot be completely segmented. The SegNet method has the worst segmentation results, which again proves that the SegNet method cannot guarantee the accuracy of oil spill segmentation for small data in the absence of datasets. This is also consistent with the results of the first two experiments. DeepLabv3+ uses the decoder to gradually return the boundary information, which improves the segmentation effect while paying attention to the boundary information. However, in this experiment, due to the lack of a training set, DeepLabv3+ only retains the boundary information and loses the internal information. MCAN outperforms U-Net++ in segmenting images with more noise interference due to the multiscale and discriminator, but MCAN still has incorrectly segmented areas, which is also consistent with the results of the first two experiments. The MTGANs method proposed in this article works well for the segmentation of oil spill images with large or small areas and high noise impact. The performance index results of the three test images in Table VII, again our judgments on the visual effects are consistent. Therefore, the MTGANs proposed in this article show excellent



Fig. 8. Experimental results of three oil spill test images from GF-3 data on five comparison methods. (a), (b), and (c) are three different test images.

TABLE VIII AVERAGE PERFORMANCE INDEX OF FIVE SEGMENTATION METHODS FOR GF-3 DATA

Methods	OA(%)	P(%)	R(%)	F1(%)	<i>K</i> (%)	M(%)
U-Net++ [20]	84.18	64.52	64.67	64.58	36.80	55.80
MODAU-Net [30]	94.29	69.05	71.01	70.02	41.62	61.66
SegNet [33]	96.34	60.02	73.33	66.21	29.50	57.54
DeepLabv3+ [34]	97.81	74.72	79.10	76.85	52.31	67.32
MCAN [54]	94.02	68.28	74.33	75.78	42.44	62.86
MTGANs	98.24	77.99	82.02	79.93	59.00	71.50

The bold values indicate that the result is optimal compared to other proposed methods.

performance in oil spill images with a large percentage of oil spill area, complex oil spill shape, small area, and high noise interference.

The average performance of the five GF-3 oil spill test images is compared, and the average performance indicators are shown in Table VIII. Overall, the MTGANs method proposed in this article has the highest Precision, F1-score, Kappa, and MIoU. Also, it is again proved that the MTGANs method proposed in this article has the best performance compared with small data segmentation performance, which is also consistent with the conclusions of Experiment 1 and Experiment 2.

G. Ablation Study

The discriminator of the classification model is composed of multiple convolutional blocks. In order to show the effect of the number of layers of convolution blocks on the recognition accuracy of oil spills and look-alike oil spills, the number of layers of convolution blocks is varied and set to 3, 4, and 5, respectively, and the classification results are tabulated in Table IX. It can be observed that the number of layers of the discriminator is not necessarily better the higher the performance. Through experimental comparison, it is found that the highest accuracy in identifying oil spills and look-alike oil spills is achieved when the number of layers of the discriminator is 4.

In the model, the discriminator D_s uses multiple layers of convolutional blocks, and the generator G_s has multiple convolutional blocks connected after the full convolutional symmetric

TABLE IX PERFORMANCE METRICS OBTAINED BY VARYING THE NUMBER OF LAYERS OF THE CLASSIFICATION MODEL DISCRIMINATOR

Layers (D_c)	Accuracy(%)
3	92.50
4	97.50
5	95.00
The bold values indic	ate that the result is

optimal compared to other proposed methods.

TABLE X Performance Metrics Obtained by Changing the Number of Layers of Generators and Discriminators

Layers (D_s)	Layers (G_s)	OA(%)	P(%)	R(%)	F1(%)	K(%)	M(%)
	2	97.33	87.74	74.99	80.87	61.84	72.44
	3	96.92	81.58	83.52	82.54	65.16	74.13
3	4	95.54	89.29	68.61	70.77	42.06	62.89
	5	91.70	71.23	76.11	73.59	48.83	64.69
	2	97.21	88.29	71.60	79.08	58.31	70.62
4	3	97.38	86.67	81.13	83.81	67.74	74.35
	4	97.27	87.85	74.25	80.48	61.07	71.93
	5	94.89	62.36	43.89	51.52	28.33	48.92
	2	97.17	88.21	70.86	78.59	57.36	70.19
	3	97.33	87.41	75.98	81.30	62.69	72.77
5	4	96.72	82.01	64.82	72.41	45.25	64.47
	5	96.19	81.59	55.97	66.40	33.49	58.97

The bold values indicate that the result is optimal compared to other proposed methods.

structure in the generator. In order to show whether the number of convolutional blocks affects the oil spill segmentation effect, the number of layers of convolutional blocks of the discriminator is set to 3, 4, and 5, and the number of layers of convolutional blocks of the generator is set to 2, 3, 4, and 5, respectively. The segmentation results are tabulated in Table X.

It can be observed that a higher number of generator layers does not necessarily lead to better performance, as a higher number of generator layers results in a longer training time and a tendency to overfit the network. A smaller number of layers does not help the network to learn the feature information. The discriminator also has an impact on the segmentation accuracy, but not as much as the generator. In contrast, by combining F1, Kappa, and MIoU coefficients for a comprehensive analysis and comparison, it is found that the segmentation network model is

 EFFECT OF BALANCING PARAMETER VALUES ON EXPERIMENTAL RESULTS

 Parameter
 Value
 OA(%)
 P(%)
 R(%)
 F1(%)
 K(%)
 M(%)

 8
 97.32
 86.80
 75.05
 80.50
 64.85
 73.82

 9
 97.35
 85.60
 78.36
 81.84
 65.74
 74.09

TABLE XI

	9	97.35	85.60	78.36	81.84	65.74	74.09
λ_1	10	97.38	86.67	81.13	83.81	67.74	74.35
	11	95.10	76.15	81.01	78.79	58.06	70.26
	8	97.34	86.94	76.12	81.76	63.59	73.22
	9	97.28	84.96	80.70	82.78	65.61	74.22
λ_2	10	97.38	86.67	81.13	83.81	67.74	74.35
	11	96.37	82.29	63.82	71.89	44.03	63.90

The bold values indicate that the result is optimal compared to other proposed methods.

optimal in terms of F1, Kappa, and MIoU coefficients when the discriminator is four layers and the generator is three layers. This also shows that setting the discriminator to 4 and the number of layers of the generator to 3 in the model is crucial to obtain a good segmentation.

In addition, the experimental comparison is carried out by adjusting the values of λ_1 and λ_2 parameters under the Sentinel-1 data. A total of 20 oil spill images are selected, and the segmentation results are analyzed by combining the evaluation index. The values of the parameters are set to 8, 9, 10, and 11, respectively. Finally, the experimental results are tabulated in Table XI. When $\lambda_1 = 10, \lambda_2 = 10$, the segmentation performance of the oil spill is optimal.

H. Running Time

The running time of the model is calculated and tested with 20 images, and the classification and segmentation models are run five times each and averaged. The average running time of the classification model is 2.10 s, and the running time of the segmentation model is 5.91 s. For each image, the time to identify the oil spill and look-alike oil spill is about 0.105 s, and the time to segment the oil spill area is 0.2955 s. Therefore, in case of an unexpected oil spill, the model can quickly identify the oil spill and look-alike oil spill and can accurately determine the extent of the oil spill. The model can promptly identify the oil spill and look-alike oil spill and accurately determine the oil spill area. The model can be used to identify oil spills and look-alike oil spills quickly and determine the extent of oil spills accurately so that the oil spill disaster can be dealt with in time and the pollution to the marine environment can be reduced.

I. Discussion

The oil spill classification experiment is carried out on the Sentinel-1 dataset. MTGANs successfully classify through the discriminator, while the generator enriches the number of adversarial samples. A lot of look-alike oil spill SAR images are fully used in model learning, significantly improving the model's discriminant ability. It is seen that accurate classification of real and look-alike oil spill images can be achieved from Fig. 5. There is no prior salient feature selection and marine physical environment element query. Therefore, MTGANs has strong discrimination ability compared with traditional machine learning methods, such as SVM, KNN, and RFC.

The segmentation experiments of oil spills are performed under three different oil spill datasets, Sentinel-1, ERS-1/2, GF-3. U-Net++, MODAU-Net, SegNet, DeepLabv3+, and MCAN methods are used for comparative analysis. The segmentation effects of all algorithms are basically consistent on the three datasets and will not be affected by different satellite imaging effects. MODAU-Net utilizes offset convolutional blocks to convert spatial information into channel information, which can accurately segment oil spill fuzzy boundaries, as shown in Fig. 7(a) and (b). But there are still a large number of areas without oil spills are predicted as oil spills in Figs. 6(b)and 8(b). The U-Net++ method resets the jump path based on U-Net and can integrate the features of different levels, so the features of different levels of the oil spill are extracted. However, U-Net++ is difficult to predict the oil spill image with large noise interference, and the oil spill area cannot be predicted in Fig. 8(b). The SegNet method is challenging to train under a small number of datasets, resulting in difficulty predicting the results, especially in the edge areas. SegNet makes unpredictable results in Figs. 7 and 8, and fuzzy boundaries in Fig. 6(a) and (b). The DeepLabv3+ method adopts dilated convolution to shrink the feature map. Then it uses the bilinear interpolation upsampling to restore the original resolution so that the operation cannot recover the lost information. The information on the oil spill is largely lost in Figs. 7(c) and 8(a). Compared with other methods, the results generated by the MCAN method are closer to the groundtruth. However, the generator employs a simple and lightweight network in the MCAN model, so it cannot fully capture the overall feature information of the oil spill. MCAN cannot maintain excellent accuracy, so misjudgments will also occur when extracting oil spill information. The MTGANs proposed in this article designs a fully convolutional symmetric structure to the generator in the segmentation stage, which can extract global to local oil spill information. It not only maintains the overall characteristics of the oil spill without losing the details but also reduces the misclassification of the oil spill from the overall to the local under the action of the discriminator. The results of three experiments are consistent with the theory analysis. Compared with other methods, MTGANs can maintain the details of oil spill edge features and accurately predict the oil spill area in Figs. 6 and 7, respectively. MTGANs still has the best accuracy facing with considerable noise interference, as shown in Fig. 8. Therefore, the proposed MTGANs can maintain excellent performance when classifying and segmenting oil spills.

V. CONCLUSION

In this article, a novel MTGANs is proposed for oil spill classification and segmentation. Different from previous works, the proposed method integrates oil spill classification and segmentation into a single framework. In the classification stage of MTGANs, the discriminator is key, and the generator is used to enrich similar samples to improve the discriminative ability of the model. In the segmentation stage, the fully convolutional symmetric structure is added to the generator to extract the oil spill information from the global to the local oil spill. It not only maintains the overall characteristics of the oil spill but also does not lose the details and dramatically reduces the error classification under the role of the discriminator. Therefore, MTGANs can not only solve the problem that oil spills and look-alike oil spills are difficult to distinguish but also solve the lack of restrictions on oil spill datasets and reduce the misjudgment rate to achieve accurate segmentation of oil spills. Experimental results on three widely used satellite datasets verify the effectiveness of the proposed method. MTGANs is superior compared with the current state-of-the-art methods.

REFERENCES

- A. Solberg, "Remote sensing of ocean oil-spill pollution," *Proc. IEEE*, vol. 100, no. 10, pp. 2931–2945, Oct. 2012.
- [2] M. M. Espeseth, C. Brekke, C. E. Jones, B. Holt, and A. Freeman, "The impact of system noise in polarimetric SAR imagery on oil spill observations," *IEEE Trans. Geosci. Remote Sens.*, vol. 58, no. 6, pp. 4194–4214, Jun. 2020.
- [3] A. H. S. Solberg, C. Brekke, and P. O. Husoy, "Oil spill detection in Radarsat and Envisat SAR images," *IEEE Trans. Geosci. Remote Sens.*, vol. 45, no. 3, pp. 746–755, Mar. 2007.
- [4] O. Garcia-Pineda et al., "Classification of oil spill by thicknesses using multiple remote sensors," *Remote Sens. Environ.*, vol. 236, 2020, Art. no. 111421.
- [5] M. Migliaccio, A. Gambardella, and M. Tranfaglia, "SAR polarimetry to observe oil spills," *IEEE Trans. Geosci. Remote Sens.*, vol. 45, no. 2, pp. 506–511, Feb. 2007.
- [6] Y. Dong, Y. Liu, C. Hu, I. R. MacDonald, and Y. Lu, "Chronic oiling in global oceans," *Science*, vol. 376, no. 6599, pp. 1300–1304, 2022.
- [7] Y. Li and Y. Zhang, "Synthetic aperture radar oil spills detection based on morphological characteristics," *Geo-Spatial Inf. Sci.*, vol. 17, no. 1, pp. 8–16, 2014.
- [8] S. Tong, X. Liu, Q. Chen, Z. Zhang, and G. Xie, "Multi-feature based ocean oil spill detection for polarimetric SAR data using random forest and the self-similarity parameter," *Remote Sens.*, vol. 11, no. 4, pp. 1–20, 2019, Art. no. 451, doi: 10.3390/rs11040451.
- [9] F. Bandiera, A. Masciullo, and G. Ricci, "A Bayesian approach to oil slicks edge detection based on SAR data," *IEEE Trans. Geosci. Remote Sens.*, vol. 52, no. 5, pp. 2901–2909, May 2014.
- [10] A. Buono, F. Nunziata, M. Migliaccio, and X. Li, "Polarimetric analysis of compact-polarimetry SAR architectures for sea oil slick observation," *IEEE Trans. Geosci. Remote Sens.*, vol. 54, no. 10, pp. 5862–5874, Oct. 2016.
- [11] F. Del Frate, A. Petrocchi, J. Lichtenegger, and G. Calabresi, "Neural networks for oil spill detection using ERS-SAR data," *IEEE Trans. Geosci. Remote Sens.*, vol. 38, no. 5, pp. 2282–2287, Sep. 2000.
- [12] Y. Li, Q. Yu, M. Xie, Z. Zhang, Z. Ma, and K. Cao, "Identifying oil spill types based on remotely sensed reflectance spectra and multiple machine learning algorithms," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 14, pp. 9071–9078, 2021.
- [13] F. M. Ghara, S. B. Shokouhi, and G. Akbarizadeh, "A new technique for segmentation of the oil spills from synthetic-aperture radar images using convolutional neural network," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 15, pp. 8834–8844, 2022.
- [14] A. Raeisi, G. Akbarizadeh, and A. Mahmoudi, "A new algorithm for oil spill detection from SAR images," in *Proc. 2nd Nat. Conf. New Marine Technologies*, 2016, pp. 1–14.
- [15] F. Sharifzadeh, G. Akbarizadeh, and Y. Seifi Kavian, "Ship classification in SAR images using a new hybrid CNN–MLP classifier," *J. Indian Soc. Remote Sens.*, vol. 47, no. 10, pp. 551–562, 2018.
- [16] G. Chen, Y. Li, G. Sun, and Y. Zhang, "Application of deep networks to oil spill detection using polarimetric synthetic aperture radar images," *Appl. Sci.*, vol. 7, p. 968, 2017.
- [17] S. Yekeen, A. Balogun, and K. Yusof, "A novel deep learning instance segmentation model for automated marine oil spill detection," *ISPRS J. Photogrammetry Remote Sens.*, vol. 167, pp. 190–200, 2020.
- [18] Q. Zhu et al., "Oil spill contextual and boundary-supervised detection network based on marine SAR images," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 5213910, doi: 10.1109/TGRS. 2021.3115492.
- [19] O. Ronneberger, P. Fischer, and T. Brox, "U-Net: Convolutional networks for biomedical image segmentation," in *Proc. Med. Image Comput. Comput.-Assist. Interv.*, 2015, pp. 234–241.

- [20] Z. Zhou, M. M. R. Siddiquee, N. Tajbakhsh, and J. Liang, "U-net++: Redesigning skip connections to exploit multiscale features in image segmentation," *IEEE Trans. Med. Imag.*, vol. 39, no. 6, pp. 1856–1867, Jun. 2020.
- [21] X. Ma, J. Xu, P. Wu, and P. Kong, "Oil spill detection based on deep convolutional neural networks using polarimetric scattering information from Sentinel-1 SAR images," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 4204713, doi: 10.1109/TGRS.2021.3126175.
- [22] L. De Laurentiis, C. E. Jones, B. Holt, G. Schiavon, and F. Del Frate, "Deep learning for mineral and biogenic oil slick classification with airborne synthetic aperture radar data," *IEEE Trans. Geosci. Remote Sens.*, vol. 59, no. 10, pp. 8455–8469, Oct. 2021.
- [23] M. Nieto-Hidalgo, A. J. Gallego, P. Gil, and A. Pertusa, "Two-stage convolutional neural network for ship and spill detection using SLAR images," *IEEE Trans. Geosci. Remote Sens.*, vol. 56, no. 9, pp. 5217–5230, Sep. 2018.
- [24] A. Basit, M. A. Siddique, and M. S. Sarfraz, "Deep learning based oil spill classification using U-net convolutional neural network," in *Proc. IEEE Int. Geosci. Remote Sens. Symp.*, 2021, pp. 3491–3494.
- [25] G. Xia, G. Liu, W. Yang, and L. Zhang, "Meaningful object segmentation from SAR images via a multiscale nonlocal active contour model," *IEEE Trans. Geosci. Remote Sens.*, vol. 54, no. 3, pp. 1860–1873, Mar. 2016.
- [26] A. Raeisi, G. Akbarizadeh, and A. Mahmoudi, "Combined method of an efficient cuckoo search algorithm and nonnegative matrix factorization of different Zernike moment features for discrimination between oil spills and lookalikes in SAR images," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 11, no. 11, pp. 4193–4205, Nov. 2018.
- [27] N. Aghaei, G. Akbarizadeh, and A. Kosarian, "GreyWolfLSM: An accurate oil spill detection method based on level set method from synthetic aperture radar imagery," *Eur. J. Remote Sens.*, vol. 55, pp. 1–18, 2022.
- [28] N. Aghaei, G. Akbarizadeh, and A. Kosarian, "OSDES-Net: Oil spill detection based on efficient-shuffle network using synthetic aperture radar imagery," *Geocarto Int.*, 2022.
- [29] X. Wang et al., "Detection of oil spill using SAR imagery based on AlexNet model," *Comput. Intell. Neurosci.*, vol. 2021, pp. 1–14, 2021.
- [30] J. Wang, J. Fan, and J. Wang, "MDOAU-Net: A lightweight and robust deep learning model for SAR image segmentation in aquaculture raft monitoring," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4504505, doi: 10.1109/LGRS.2022.3147355.
- [31] W. Li, Y. Gao, M. Zhang, R. Tao, and Q. Du, "Asymmetric feature fusion network for hyperspectral and SAR image classification," *IEEE Trans. Neural Netw. Learn. Syst.*, to be published, doi: 10.1109/TNNLS.2022.3149394.
- [32] J. Wang, W. Li, Y. Gao, M. Zhang, R. Tao, and Q. Du, "Hyperspectral and SAR image classification via multiscale interactive fusion network," *IEEE Trans. Neural Netw. Learn. Syst.*, to be published, doi: 10.1109/TNNLS.2022.3171572.
- [33] V. Badrinarayanan, A. Kendall, and R. Cipolla, "SegNet: A deep convolutional encoder-decoder architecture for image segmentation," *IEEE Trans. Pattern Anal. Mach. Intell.*, vol. 39, no. 12, pp. 2481–2495, Dec. 2017.
- [34] L. Chen, Y. Zhu, G. Papandreou, F. Schroff, and H. Adam, "Encoderdecoder with atrous separable convolution for semantic image segmentation," in *Proc. Comput. Vis.*, 2018, pp. 833–851.
- [35] Y. Zhang, W. Li, M. Zhang, S. Wang, R. Tao, and Q. Du, "Graph information aggregation cross-domain few-shot learning for hyperspectral image classification," *IEEE Trans. Neural Netw. Learn. Syst.*, to be published, doi: 10.1109/TNNLS.2022.3185795.
- [36] C. Finn, P. Abbeel, and S. Levine, "Model-agnostic meta-learning for fast adaptation of deep networks," in *Proc. 34th Int. Conf. Mach. Learn.*, 2017, vol. 70, pp. 1126–1135.
- [37] D. A. Alajaji and H. Alhichri, "Few shot scene classification in remote sensing using meta-agnostic machine," in *Proc. Conf. Data Sci. Mach. Learn. Appl.*, 2020, pp. 77–80.
- [38] M. Rao, P. Tang, and Z. Zhang, "Spatial-spectral relation network for hyperspectral image classification with limited training samples," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 12, no. 12, pp. 5086–5100, Dec. 2019.
- [39] B. Wang, Z. Wang, X. Sun, H. Wang, and K. Fu, "DMML-Net: Deep metametric learning for few-shot geographic object segmentation in remote sensing imagery," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 5611118, doi: 10.1109/TGRS.2021.3116672.
- [40] Z. Huang, Z. Pan, and B. Lei, "Transfer learning with deep convolutional neural network for SAR target classification with limited labeled data," *Remote Sens.*, vol. 9, no. 9, pp. 2072–4292, 2017.

- [41] M. Rostami, S. Kolouri, E. Eaton, and K. Kim, "Deep transfer learning for few-shot SAR image classification," *Remote Sens.*, vol. 11, no. 11, pp. 2072–4292, 2019.
- [42] X. Yu, X. Wu, C. Luo, and P. Ren, "Deep learning in remote sensing scene classification: A data augmentation enhanced convolutional neural network framework," *GIScience Remote Sens.*, vol. 54, pp. 1–18, 2017.
- [43] I. J. Goodfellow et al., "Generative adversarial nets," in Proc. 27th Int. Conf. Neural Inf. Process. Syst., 2014, pp. 2672–2680.
- [44] M. Mirza and S. Osindero, "Conditional generative adversarial nets," 2014, arXiv:1411.1784.
- [45] A. Radford, L. Metz, and S. Chintala, "Unsupervised representation learning with deep convolutional generative adversarial networks," 2015, arXiv: 1511.06434.
- [46] A. Odena, "Semi-supervised learning with generative adversarial networks," 2016, arXiv:1606.01583.
- [47] M. Arjovsky, S. Chintala, and L. Bottou, "Wasserstein GAN," 2017, arXiv:1701.07875.
- [48] I. Gulrajani, F. Ahmed, M. Arjovsky, V. Dumoulin, and A. Courville, "Improved training of Wasserstein GANs," in *Proc. 31st Int. Conf. Neural Inf. Process. Syst.*, 2017, pp. 5769–5779.
- [49] Z. Zhang, L. Yang, and Y. Zheng, "Translating and segmenting multimodal medical volumes with cycle- and shape-consistency generative adversarial network," in *Proc. IEEE Conf. Comput. Vis. Pattern Recognit.*, 2018, pp. 9242–9251.
- [50] X. Pan, J. Zhao, and J. Xu, "Conditional generative adversarial networkbased training sample set improvement model for the semantic segmentation of high-resolution remote sensing images," *IEEE Trans. Geosci. Remote Sens.*, vol. 59, no. 9, pp. 7854–7870, Sep. 2021.
- [51] A. Ma, N. Yu, Z. Zheng, Y. Zhong, and L. Zhang, "A supervised progressive growing generative adversarial network for remote sensing image scene classification," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 5618818, doi: 10.1109/TGRS.2022.3151405.
- [52] J. Feng, H. Yu, L. Wang, X. Cao, X. Zhang, and L. Jiao, "Classification of hyperspectral images based on multiclass spatial–spectral generative adversarial networks," *IEEE Trans. Geosci. Remote Sens.*, vol. 57, no. 8, pp. 5329–5343, Aug. 2019.
- [53] X. Yu, H. Zhang, C. Luo, H. Qi, and P. Ren, "Oil spill segmentation via adversarial *f*-divergence learning," *IEEE Trans. Geosci. Remote Sens.*, vol. 56, no. 9, pp. 4973–4988, Sep. 2018.
- [54] Y. Li, X. Lyu, A. C. Frery, and P. Ren, "Oil spill detection with multiscale conditional adversarial networks with small-data training," *Remote Sens.*, vol. 13, no. 12, pp. 2072–4292, 2021.
- [55] L. Samaniego, A. Bardossy, and K. Schulz, "Supervised classification of remotely sensed imagery using a modified K-NN technique," *IEEE Trans. Geosci. Remote Sens.*, vol. 46, no. 7, pp. 2112–2125, Jul. 2008.



Jianchao Fan (Member, IEEE) received the B.S. and Ph.D. degrees in control science and engineering from the School of Control Science and Engineering, Dalian University of Technology, Dalian, China, in 2007 and 2012, respectively.

During 2013–2016, he was a Postdoctoral Fellow jointly cultivated by The Chinese University of Hong Kong and Dalian University of Technology, and during 2018–2019, he was a Visiting Scholar with the Department of Computer Science and Engineering, Washington University in St. Louis, St. Louis, MO,

USA. Since 2019, he has been a Professor with the Department of Ocean Remote Sensing, National Marine Environmental Monitoring Center, Dalian, China. He has authored or coauthored three books and more than 80 scientific papers in international journals and conference proceedings, such as the IEEE TRANSACTIONS ON FUZZY SYSTEMS, IEEE TRANSACTIONS ON NEURAL NETWORKS AND LEARNING SYSTEMS, IEEE TRANSACTIONS ON GEOSCIENCE AND REMOTE SENSING, IEEE JOURNAL OF SELECTED TOPICS IN APPLIED EARTH OBSERVATIONS AND REMOTE SENSING, IEEE GEOSCIENCE AND REMOTE SENSING LETTERS, PATTERN RECOGNITION. His research interests include marine SAR image processing, polarimetric SAR, deep learning neural networks, and heuristic swarm intelligent optimization.

Dr. Fan is a Program Chair, Workshop Chair, Publicity Chair, and Local Arrangement Chair of several international conferences, such as the International Symposium on Neural Networks, International Conference on Information Society and Technology, and International Conference on Intelligent Control and Information Processing.



Chuan Liu received the B.S. degree in software engineering from the School of Software, Zhongyuan University of Technology, Zhengzhou, China, in 2020. He is currently working toward the M.S. degree in computer science and technology with the School of Information, Shanghai Ocean University, Shanghai, China.

His research interests include generative adversarial networks and ocean remote sensing image processing.